Author's response to reviews

Title: Asymptomatic schwannoma of the heart

Authors:

Sarah A Early (earlysarah@gmail.com)
Jonathan McGuinness (mcguinness@rcsi.ie)
John Galvin (jgalvin@mater.ie)
Maria Kennedy (mkennedy@mater.ie)
John Hurley (jhurley@mater.ie)

Version: 4 Date: 3 January 2007

Author's response to reviews:

Title - Changed from 'Schwannoma' to 'schwannoma' without the capital, updated submission system title.

Deleted text 'authors:' from the title page.

Removed Word count - from the manuscript.

Removed subject category from the title page.

Removed Keywords section

Changed Clinical Summary to Background.

Moved the Figure legend section to after the References.